Author's response to reviews

Title: Validation of human papillomavirus genotyping by signature DNA sequence analysis

Authors:

Sin Hang Lee (sinhang.lee@milfordhospital.org)
Veronica S Vigliotti (veronica.vigliotti@milfordhospital.org)
Jessica S Vigliotti (09_jvigliotti@stagweb.fairfield.edu)
Suri Pappu (suri.pappu@milfordhospital.org)

Version: 6 Date: 18 May 2009

Author's response to reviews:

Dear Editorial Production Team:

Per your request, the following formatting changes have been made to the manuscript:

Major revision: Reference [10] has been cited in the correct position.
Minor revision: Title-all unnecessary captitals have been corrected.
Coloured text- the coloured text has been removed.

Please inform the undersigned if further reformatting is needed.

Thank you.

Sincerely,

Sin Hang Lee, MD
Corresponding author